Author's response to reviews

Title: An experimental study of sexual function improving effect of Myristica fragrans(nutmeg)

Authors:
- Tajuddin (tajuddin_amu@rediffmail.com)
- Shamshad - Ahmad (shamshad_2003@rediffmail.com)
- Abdul - Latif (latifabdul@rediffmail.com)
- Iqbal Ahmad - Qasmi (qasmiqbal@rediffmail.com)
- Kunwar Mohammad Yusuf - Amin (kmya55@yahoo.com)

Version: 3  Date: 7 July 2005

Author's response to reviews:

Title: An experimental study of sexual function improving effect of Myristica fragrans Houtt.(nutmeg)
MS:166844000589438

To,

Dr. Sally Dennison
Assistant Editor, Biomed central journals.

Dear Editor,
I am submitting herewith a revised version of my manuscript along with a detail of formatting changes made in the manuscript as required by you for your consideration.

;Formatting changes;

Author list: The list of authors has now been written exactly as they are in the submission system.
--- The language corrections have been made in the revised manuscript(version 3) wherever required with the help of a colleague.

---- Equal contributors: These have been indicated on the title page of the revised manuscript.

with best regards,
Tajuddin dDept. of Ilmul Advia ,AMU,Aigarh,INDIA.